Author's response to reviews

Title: Glucagon-like peptide analogues for type 2 diabetes mellitus: systematic review and meta-analysis

Authors:

Deepson S Shyangdan (d.s.shyangdan@abdn.ac.uk)
Pamela L Royle (p.royle@abdn.ac.uk)
Christine Clar (clar@cc-archie.de)
Pawana Sharma (p.sharma@abdn.ac.uk)
Norman R Waugh (n.r.waugh@abdn.ac.uk)

Version: 8 Date: 9 December 2010

Author's response to reviews: see over
Thursday, 9 December 2010

BioMed Central
Endocrine Disorders

Dear Editors,

Please find attached a revised version of our paper. Our changes are tabulated below.

Minor revisions
- Title: Hyphen added between glucagon and like.
- Title: All capitals removed except initial G.
- Title page: ‘pop health’ capitalised to ‘Population Health’.
- Address changed to: Section of Population Health, Medical School Buildings, Foresterhill, University of Aberdeen, Aberdeen, AB25 2ZD.
- In abstract and methods section page 5: Cochrane library- L capitalised.
- ‘either’ deleted from page 6 paragraph 2.
- Page 8, paragraph 1, outcomes: ‘proportion of patients with HbA1c ≤ 7% …’ added.
- Full stop added after exenatide: page 23, fifth line from bottom).
- Kendall paper cited [no. 46 in reference list]: page 29, last line, first paragraph.
- Manuscript was checked thoroughly for any typographical errors. This is our final change to the manuscript.

Prepared by:

Deepson Shyangdan, research assistant Aberdeen Health Technology Assessment Group and approved by all authors.